Author's response to reviews

Title: Noninvasive Fetal Trisomy Test (NIFTY) An Advanced Noninvasive Prenatal Diagnosis Methodology for Fetal Autosomal and Heterosomal Aneuploidies

Authors:

Fuman Jiang (jiangfuman@genomics.org.cn)
Jinghui Ren (szrenjh@126.com)
Fang Chen (fangchen@genomics.org.cn)
Yuqiu Zhou (zhzhouyq@126.com)
Jiansheng Xie (jianshengxie2000@yahoo.com.cn)
Shan Dan (sd_bao@sina.com)
Yue Su (suyue501@sohu.com)
Jianhong Xie (kdsxjh@yahoo.com.cn)
Baomin Yin (bbyin@163.com)
Wen Su (wensu_xy@hotmail.com)
Huakun Zhang (kunkun21cen@163.com)
Wei Wang (wangw@genomics.org.cn)
Xianghua Chai (chaixianghua@genomics.org.cn)
Linhua Lin (linlinhua8@sina.com)
Hui Guo (guohui600@163.com)
Qiyun Li (qiyun53@sina.com)
Peipei Li (lipipepei@genomics.org.cn)
Yuying Yuan (yuanyuying@genomics.org.cn)
Xiaoyu Pan (panxiaoyu@genomics.org.cn)
Yihan Li (liyihan@genomics.org.cn)
Lifu Liu (liulifu@genomics.org.cn)
Huifei Chen (chenhuifei@139.com)
Zhaoling Xuan (xuanzl@genomics.org.cn)
Shengpei Chen (chenshengpei@genomics.org.cn)
Chunlei Zhang (zhangchunlei@genomics.org.cn)
Hongyun Zhang (zhanghy@genomics.org.cn)
Zhongming Tian (tianzhm@genomics.org.cn)
Zhengyu Zhang (zhangzhengyu@genomics.org.cn)
Hui Jiang (jianghui@genomics.org.cn)
Lijian Zhao (zhaolijian@genomics.org.cn)
Weimou Zheng (zhengwm@genomics.org.cn)
Songgang Li (lisg@genomics.org.cn)
Yingrui Li (liyr@genomics.org.cn)
Jun Wang (wangj@genomics.org.cn)
Jian Wang (wangjian@genomics.org.cn)
Xiuqing Zhang (zhangxq@genomics.org.cn)

Version: 4 Date: 2 May 2012
Author’s response to reviews: see over
May. 2, 2012

Dear Dr. Patrick Tan,

Thank you for giving us the opportunity to revise the manuscript entitled "Noninvasive Fetal Trisomy Test (NIFTY): An Advanced Noninvasive Prenatal Diagnosis Methodology for Fetal Autosomal and Sex chromosome Aneuploidies". We have read your comments carefully, and have made necessary changes accordingly.

We have revised the manuscript according to comments of the reviewers. A detailed response to the comments and suggestions from each reviewer is listed below in the following pages. We appreciate the patience and help from editors.

Please kindly consider the revised version for publication in your Journal. If you have any questions, please feel free to contact us.

Yours sincerely,

Xiuqing.Zhang, Ph.D.

BGI-shenzhen
Beishan Industrial Zone, ShenZhen 518083, Guangdong P. R. China.
Phone and fax number: 86 775 2527 3884
Email address: zhangxq@genomics.org.cn